Author's response to reviews

Title: The QICKD study protocol: a cluster randomised trial to compare quality improvement interventions to lower systolic BP in chronic kidney disease (CKD) in primary care [ISRCTN56023731]

Authors:

Simon de Lusignan (slusigna@sgul.ac.uk)
Hugh Gallagher (Hugh.Gallagher@epsom-sthelier.nhs.uk)
Tom Chan (tchan@sgul.ac.uk)
Nicki Thomas (N.M.Thomas@city.ac.uk)
Jeremy van Vlymen (jvanvlym@sgul.ac.uk)
Michael Nation (michaelnation@kidneyresearchuk.org)
Neerja Jain (NeerjaJain@kidneyresearchuk.org)
Aumran Tahir (mtahir@nhs.net)
Elizabeth du Bois (Elizabeth.Dubois@wpct.nhs.uk)
Iain Crinson (icrinson@sgul.ac.uk)
Nigel Hague (njhmq@hotmail.co.uk)
Fiona Reid (freid@sgul.ac.uk)
Kevin Harris (Kevin.Harris@uhl-tr.nhs.uk)

Version: 8 Date: 31 May 2009

Author's response to reviews:

Dear Editor,

We have made further changes requested by the copy editor.

We have increased the size of the diagrams submitted as Word files and changed the orientation to landscape as we can't otherwise reduce the size of figures in Word without changing to non-standard page sizes. We have also made the "tables" / Text boxes appendices as requested - however if there is a technical way they can stay nearer their original place in the text we would prefer it.

Best wishes,

Simon de Lusignan, on behalf of the authors.